Author's response to reviews

Title: Congenital Rubella Syndrome in Iran

Authors:

Jila Sadighi (jilasadigh@yahoo.com)
Hassan Eftekhar (info@ihsr.ac.ir)
Kazem Mohammad (info@ihsr.ac.ir)

Version: 2 Date: 14 November 2004

Author's response to reviews:

Dear Editor

The changes are as following:
1- revise article to standard format of the journal.
2- summarize analysis
3- Two tables (exclude one table and its details in the analysis and discussion)

Best regards,
Dr. Jila Sadighi